

Citation: Ribeiro JM, Silva PA, Costa-Silva HM, Santi AMM, Murta SMF (2024) Deletion of the lipid droplet protein kinase gene affects lipid droplets biogenesis, parasite infectivity, and resistance to trivalent antimony in *Leishmania infantum*. PLoS Negl Trop Dis 18(1): e0011880. https://doi.org/ 10.1371/journal.pntd.0011880

Editor: Albert Descoteaux, Institut national de la recherche scientifique, CANADA

Received: April 15, 2023

Accepted: December 21, 2023

Published: January 18, 2024

Copyright: © 2024 Ribeiro et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its <u>Supporting Information</u> files.

Funding: This investigation received financial support from the following agencies: Fundação de Amparo à Pesquisa do Estado de Minas Gerais (FAPEMIG: APQ-02816-21; BPD-00657-22; RED-00104-22; UGA/FAPEMIG APQ-04382-16-D to SMFM), Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq 304158/2019-4 to RESEARCH ARTICLE

Deletion of the lipid droplet protein kinase gene affects lipid droplets biogenesis, parasite infectivity, and resistance to trivalent antimony in *Leishmania infantum*

Juliana Martins Ribeiro[©], Paula Alves Silva[©], Héllida Marina Costa-Silva, Ana Maria Murta Santi, Silvane Maria Fonseca Murta_©*

Grupo Genômica Funcional de Parasitos, Instituto René Rachou, Fiocruz Minas, Belo Horizonte, Minas Gerais, Brazil

• These authors contributed equally to this work.

* silvane.murta@fiocruz.br

Abstract

The Lipid Droplet Protein Kinase (LDK) facilitates lipid droplet (LD) biogenesis, organelles involved in various metabolic and signaling functions in trypanosomatids. As LDK's function has not been previously explored in Leishmania spp., we utilized CRISPR/Cas9 technology to generate LDK-knockout lines of Leishmania infantum to investigate its role in this parasite. Our findings demonstrate that LDK is not an essential gene for L. infantum, as its deletion did not impede parasite survival. Furthermore, removing LDK did not impact the growth of promastigote forms of L. infantum lacking LDK. However, a noticeable reduction in LDs occurred during the stationary phase of parasite growth following LDK deletion. In the presence of myriocin, a LD inducer, LDK-knockout parasites displayed reduced LD abundance during both logarithmic and stationary growth phases compared to control parasites. Moreover, an infection analysis involving THP-1 cells revealed that 72 h post-infection, LDKknockout L. infantum lines exhibited fewer infected macrophages and intracellular amastigotes than control parasites. LDK-knockout L. infantum lines also displayed 1.7 to 1.8 -fold greater resistance to trivalent antimony than control parasites. There were no observed alterations in susceptibility to amphotericin B, miltefosine, or menadione in LDK-knockout L. infantum lines. Our results suggest that LDK plays a crucial role in the biogenesis and/or maintenance of LDs in L. infantum, as well as in parasite infectivity and resistance to trivalent antimony.

Author summary

Lipid droplets (LDs) are organelles with diverse cell metabolism and signaling roles. Lipid droplet protein kinase (LDK) is responsible for LD formation in trypanosomatids, but its function in *Leishmania* spp. remains unknown. In this study, we demonstrate that deleting the *LDK* gene in *Leishmania infantum* has no impact on parasite growth. However, it

SMFM; 152027/2022-0 to JMR; 150691/2023-8 to HMCS), Programa INOVA FIOCRUZ - Fundação Oswaldo Cruz (VPPCB-007-FIO-18-2-94 to SMFM); Convênio Fiocruz-Institut Pasteur-USP (no grant number to SMFM), Programa de Pósgraduação em Ciências da Saúde do Instituto René Rachou/FIOCRUZ (to SMFM) and Coordenação de Aperfeiçoamento de Pessoal de Nível Superior -Brasil (CAPES) (Finance Code 001 to PAS and AMMS). The funders had no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

reduces the amount of LDs during the stationary phase of parasite growth. When exposed to myriocin, known to induce LD production, we observed a decrease in LDs during both the logarithmic and stationary phases of *LDK*-knockout parasites. Additionally, *LDK*-deficient parasites exhibit reduced infectivity and increased resistance to trivalent antimony. In summary, our findings suggest that LDK plays a crucial role in the formation and/or maintenance of LDs in *L. infantum*.

Introduction

Leishmaniasis encompasses a complex of diseases caused by over 21 distinct species of protozoan parasites belonging to the genus *Leishmania*. It is transmitted to humans through the bites of infected female sand flies [1]. This disease predominantly affects vulnerable populations across more than 98 countries, spanning tropical and subtropical regions in the Americas, Africa, Asia, Europe, and Oceania. Over one billion people live at risk of infection in these endemic areas, resulting in approximately 0.7 to 1 million new cases annually [1,2].

Leishmaniasis manifests in two primary clinical forms: cutaneous leishmaniasis (CL), characterized by cutaneous and/or mucosal lesions, and visceral leishmaniasis (VL) [3]. VL, caused by *Leishmania (Leishmania) donovani* in Asia and Africa and *Leishmania (Leishmania) infantum* in the Mediterranean Basin, the Middle East, Central Asia, South America, and Central America, represents the most severe form of the disease and can be fatal without prompt treatment [3,4]. The progression of infection and clinical manifestations of leishmaniasis are influenced by both parasite-related factors (species, strain, life stage, initial parasitic load) and hostrelated factors (immune and nutritional status, age, presence of co-morbidities) [2–4]. Currently, no vaccines are available, and the absence of effective vector control programs underscores the reliance on chemotherapy as the primary strategy for managing all forms of leishmaniasis. The drugs used for leishmaniasis treatment include pentavalent antimonials, amphotericin B formulations, and miltefosine. However, the emergence of drug-resistant parasite strains and undesirable drug side effects contribute to treatment failures [5].

Given this context, developing novel therapeutic approaches against leishmaniasis is paramount, with the lipid droplet protein kinase (LDK) emerging as a promising molecular target. Initial investigations into LDK in *Trypanosoma brucei* revealed its involvement in lipid droplet (LD) formation and biogenesis [6]. LDs, also referred to as lipid bodies, adiposomes, or oil bodies in plants, are cytoplasmic organelles composed of a hydrophobic core containing triglycerides and cholesteryl esters enveloped in a phospholipid monolayer, cholesterol, and a set of associated proteins [7–9]. LDs serve various roles in cellular metabolism, homeostasis, and signaling [7–9] and are present in virtually all cell types and organisms, including protozoan parasites [10].

The mechanisms governing LD biogenesis in trypanosomatids remain incompletely elucidated and may involve multiple proteins [11]. To date, only LDK [6] and Lipin (TbLpn) [12] have been identified as enzymes participating in LD formation in *T. brucei*. In mammalian cells, the PAT protein family, which includes Perilipin, Adipose differentiation-related protein (ADRP), and Tail-interacting protein 47 kDa (TIP47), plays a structural role in LD formation [11,13]. Genes related to PAT proteins have not yet been identified in *Leishmania* spp. genomes, making LDK the only known protein with a comparable function [11].

LDK is primarily located at the periphery of LDs [6,14], and it possesses all the motifs and residues necessary for protein kinase activity, followed by a 24-amino acid transmembrane domain similar to *T. brucei* [6]. The kinase activity of LDK is mediated through

autophosphorylation. Moreover, studies have demonstrated that *T. brucei* mutant parasites with reduced *LDK* expression, generated via RNA interference, displayed minimal growth defects but exhibited a 90% reduction in LD formation [6]. LDK localization was assessed in *Leishmania mexicana* using CRISPR/Cas9 gene editing, employing labeling with the fluorescent protein mNeonGreen [14]. As observed in *T. brucei*, LDK in *L. mexicana* is also found in LDs. Nevertheless, the specific function of LDK in *Leishmania* spp. remains unknown.

To unravel the role of LDK in *L*. infantum, we employed the CRISPR/Cas9 approach to disrupt the *LDK* gene. We assessed the phenotypes of these mutant parasites in terms of growth rate, LD production, infectivity in THP-1 macrophages, and susceptibility to trivalent antimony (Sb^{III}), amphotericin B, miltefosine, and menadione.

Methods

Parasites culture conditions

Leishmania (Leishmania) infantum RPV (MHOM/BR/2002/LPC-RPV) promastigotes were cultured at 26°C in M199 medium (Gibco, Waltham, MA, USA) supplemented with the following components: 40 mM HEPES (pH 7.4), 5 µg/mL hemin, 2 µg/mL biopterin, 1 µg/mL biotin, 2 mM L-glutamine, 500 U/mL penicillin, 50 µg/mL streptomycin, and 10% inactivated fetal bovine serum. To maintain the cultures, we conducted biweekly passages, inoculating 1×10^6 parasites for each 5 mL of medium. Unless specified otherwise, all experiments were conducted using promastigotes in the logarithmic growth phase.

Generation of LDK L. infantum knockout

To generate *LDK* (LINF_280026600) null mutants, we employed the CRISPR/Cas9 method, following the procedure outlined in a previous study [15]. Plasmid pTB007, containing hygromycin as a resistance marker, was utilized to express SpCas9 and T7RNAP. Parasites carrying this plasmid, successfully expressing Cas9, were subjected to transfection with donor DNAs and guide RNA templates chosen using the LeishGEdit tool. The sgRNA templates were generated through polymerase chain reaction (PCR) employing G00 primer. Donor DNAs aimed at deleting the *LDK* gene were amplified via PCR, using specific primer pairs along with plasmids pTNEO_v1 and pTBlast_v1 as templates. These plasmids contain sequences that confer resistance to the antibiotics neomycin (NEO) and blasticidin (BLAST), respectively. Promastigote forms were transfected [16] with equimolar amounts of sgRNA and donor DNAs, as previously described [15]. Subsequently, parasites were selected using 40 µg/mL of NEO and 30 µg/mL of BLAST. Clones were further selected in semi-solid M199 medium with the addition of selective drugs. However, it's important to note that all experiments aimed at determining parasite phenotypes were conducted without the use of selection drugs. All primers used to construct the plasmids and DNA fragments are listed in S1 Table.

Add-back parasites

The *LDK* coding sequence was cloned into the pIR1_SAT vector to generate the add-back parasites. After ligation, the resulting plasmids were incubated with *Escherichia coli* TOP10F' (Invitrogen, Waltham, MA, USA) at 4°C for 30 minutes and then at 42°C for 45 seconds using a dry bath. Following this, the bacteria were incubated at 37°C for 1 hour to induce expression of the ampicillin resistance gene and subsequently plated on solid LB medium containing ampicillin (Sigma, St. Louis, MO, USA). Clones possessing the correct LDK sequence, confirmed via colony PCR, were subjected to digestion with the restriction enzymes *Bgl*II (Bio-Labs, Durham, NC, USA), *Sac*II (Promega, Madison, WI, USA), and *Eco*RV (Promega) to verify the orientation of the gene within the plasmid. Subsequently, 100 μ g of the pIR1_-SAT_LDK plasmid was linearized with using the restriction enzyme *Swa*I (BioLabs) and precipitated using sodium acetate and isopropanol. Transfection of Δ LDK mutant promastigotes with the pIR1_SAT_LDK plasmid was conducted following the established protocol [16], and the parasites were selected using 100 μ g/mL of nourseothricin sulfate (Sigma).

PCR reactions

To screen for clones and add-back parasites, we extracted genomic DNA from the parasites using the phenol/chloroform method. The complete *LDK* coding sequence was then amplified via PCR, and the PCR product was analyzed using agarose gel electrophoresis. To evaluate *LDK* transcript levels, we extracted total RNA using the TRIzol (Invitrogen)/chloroform method. Subsequently, total RNA was treated with DNase I (Ambion, Waltham, MA, USA), and cDNA synthesis was carried out using Superscript II reverse transcriptase (Invitrogen) following the manufacturer's instructions.

To quantify the number of copies and the transcript levels of the LDK gene, we diluted all genomic DNA and cDNA samples to 100 ng/ μ L and used them in the RT-qPCR amplification reaction. The amplification was performed with 1X SYBR GREEN master mix (Applied Biosystems, Waltham, MA, USA) and the specific primers listed in S1 Table. DNA polymerase I served as a normalizer and an endogenous control gene. All reactions were conducted in triplicate in a 96-well plate. The amplification process involved an initial cycle of 50°C for 2 minutes and 95°C for 10 minutes, followed by 40 cycles of 95°C for 15 seconds and 60°C for 1 minute each, and concluded with a final cycle of 95°C for 15 seconds, 60°C for 15 seconds, and 95°C for 15 seconds. Relative expression levels were calculated using the 2^{- $\Delta\Delta$ CT} method [17], normalized to the constitutive gene DNA polymerase I. We performed three independent biological replicates for each parasite line, with each replicate conducted in triplicate.

Growth curve of promastigotes forms

L. infantum promastigotes $(1 \times 10^6 \text{ parasites/mL})$ were maintained in M199 medium at 26°C, and the number of parasites was determined daily using theZ1 Coulter Particle Counter (Beckman Coulter, Brea, CA, USA). Three independent experiments were performed in triplicates.

Nile red and Bodipy 493/503 staining and quantification

L. infantum promastigotes $(3 \times 10^6$ parasites) in both the logarithmic and stationary growth phases were cultured for 24 h, with or without 1.5 µM myriocin (Sigma-Aldrich) [6,18]. The parasites were then washed twice with phosphate-buffered saline and stained with Nile red (1.5 µg/mL) (Sigma-Aldrich) or Bodipy 493/503 (5 µg/mL) (Thermo Fisher Scientific) for 30 min at 25°C under dark. To assess the quantity of lipid droplets within the parasites, we acquired a total of 30,000 events/sample using the BD FACSCalibur flow cytometer (BD Biosciences, Franklin Lakes, NJ, USA), and the data were analyzed using Flow Jo software (Flow Cytometry Analysis Software, version 10) [19]. For confocal microscopy, a portion of each cell suspension was incubated on a glass coverslip pre-coated with 0.1% poly-L-lysine for 10 min and subsequently fixed in 4% formaldehyde for an additional 10 min at 25°C, in dark. The glass coverslips were washed thrice with phosphate-buffered saline and incubated with Pro-Long Gold antifade reagent. We captured images using a Zeiss LSM 880 confocal microscope (Oberkochen, Germany) with a 63x objective and numerical aperture of 1.4. The acquired images were analyzed using the Zeiss Zen 3.8 microscopy software. Five independent experiments were performed in duplicate.

THP-1 derived macrophages infection

THP-1 cells (human monocytes lineage) were cultured in RPMI-1640 medium supplemented with 2 mM glutamine, 100 U/mL penicillin and 100 μ g/mL streptomycin antibiotics, and 10% fetal bovine serum, and maintained at 37°C and 5% CO₂. For facilitating the differentiation of monocytes into macrophages, the cells were incubated with 20 ng/mL of phorbol myristate acetate in 24-well plates on glass coverslips, at 37°C and 5% CO₂. After 72 h, the macrophages were infected with *Leishmania* cultures on the second day of the stationary phase (10 parasites per macrophage) for 3 h. The parasites that failed to infect the macrophages were washed away, and the infected macrophages were incubated for 72 h in RPMI-1640 medium. The infectivity of *L. infantum* mutants and add-backs was evaluated immediately after the 3 h incubation period as well as after 72 h. Slides were stained with Rapid Panoptic (Laborclin, Pinhais, PR, Brazil), photographed, and infection was assessed by counting the number of infected macrophages and intracellular amastigotes using the ImageJ software (National Institutes of Health, Bethesda, MD, USA). Three independent experiments were performed in duplicate.

Effective concentration (EC₅₀) assays

The susceptibility of the parasites to different drugs was assessed using $2 \times 10^5 L$. infantum promastigotes/mL incubated in 1 mL of M199 medium containing different drug concentrations ranging from 50 to 250 μ M of Sb^{III}, 0.05 to 0.6 μ M of amphotericin B, 5.0 to 40 μ M of miltefosine, and 1.0 to 8 μ M of menadione. After 48 h of incubation, the number of parasites grown in the absence and presence of each drug was determined by using the Z1 Coulter Particle Counter (Beckman Coulter). The values of the effective concentration necessary to reduce growth by 50% (EC₅₀) were obtained from three independent experiments in triplicate, using the linear interpolation method [20].

Statistical analysis

Data were analyzed using GraphPad Prism v.8.2.0 software (San Diego, CA). Ordinary oneway ANOVA or two-way ANOVA test with Dunnett's post-test were used to compare Cas9 and mutant parasites. Data are presented as mean \pm standard error of the mean. Statistically significant differences were considered for *p* lower than 0.05.

Results

Generation and characterization of LDK null mutants

The genome of the JPCM5 strain of *L. infantum* harbors a single copy of the *LDK* gene (TritrypDB accession number LINF_280026600), located on chromosome 28. This gene spans 2631 base pairs and encodes a protein comprising 876 amino acids. To generate *LDK* knockout cell lines, we employed the CRISPR/Cas9 methodology, following established protocols [15]. Confirmation of the correct integration of both the BLAST and neomycin cassettes into the *L. infantum* genome was achieved through PCR analysis in all tested mutant *L. infantum* clones (Fig 1A and 1B). Additionally, PCR was employed to verify the deletion of *LDK* in these parasites, successfully establishing *LDK*-null mutants. These generated clones were designated as Δ LDK cl1 and Δ LDK cl2 (Fig 1C). The reintroduction of the *LDK* gene was confirmed via PCR, and the resulting add-back parasites were labeled as cl1 + add-back and cl2 + add-back (Fig 1C). As depicted in Fig 1D, the number of *LDK* gene copies indicate complete deletion in the knockout clones, which was subsequently reinstated in the add-back parasites. Fig 1E illustrates the transcript levels of the *LDK* gene; in *LDK*-knockout parasites, *LDK* expression was absent, while the add-back parasites exhibited a restored genotype.



Fig 1. Characterization of *Leishmania infantum* Δ LDK knockout clones generated using CRISPR/Cas9 and (LDK) add-back parasites. (A) The correct integration of the resistance markers BLAST (blasticidin) (1069 bp) and (B) NEO (neomycin) (1327 bp) was evaluated using PCR by annealing a primer in a 3'UTR region adjacent to the cassette (primer P1) and another primer annealed within each resistance marker sequence (primers P2 or P3). (C) The complete *LDK* coding sequence (2631 bp) was amplified using PCR with primers P4 and P5. RT-qPCR was performed with specific primers and the number of copies of the *LDK* gene were analyzed using the 2'- Δ CT method and normalized to the constitutive gene DNA polymerase I. (D) Number of copies of the *LDK* gene. (E) Transcript levels of *LDK*. One-way analysis of variance (ANOVA) with Dunnett's post hoc test was used to compare control parasite (Cas9) and add-back parasites. *represents significant differences in relation to the control parasite (Cas9) (*** *p* < 0.001). MW: molecular weight; NC: negative control; bp: base pair.

https://doi.org/10.1371/journal.pntd.0011880.g001

LDK is not essential for *L. infantum* survival, and its deletion did not alter parasite growth

The growth of the promastigote forms of *L. infantum* parasites expressing Cas9 and Δ LDK mutant clones cl1 and cl2 was determined by counting the parasites every 24 h. No difference in growth was observed among parasites expressing Cas9, Δ LDK mutant parasites, as well the add-back parasites (S1 Fig). The results demonstrated that *LDK* is not an essential gene for *L. infantum*, and its deletion did not alter parasite growth.

LDK knockout reduced LD production in the stationary phase

To evaluate the effect of *LDK* deletion on the formation of LDs in *L. infantum*, we quantified LDs using flow cytometry based on the geometric mean fluorescence intensity (gMFI) of Nile red (Fig 2) and Bodipy 493/503 (S2 Fig) staining in the logarithmic and stationary phases of parasite growth. No significant difference was observed in the amount of LDs in the knockout parasites in the logarithmic phase when compared to that of the Cas9-expressing controls, and the profile did not change with the reintroduction of the *LDK* gene into the add-back parasites (Figs 2A and S2A). However, in the stationary phase, a reduction in the amount of LDs in *LDK*-knockout parasites (Figs 2A and S2A) was observed. Reintroduction of the *LDK* gene into knockout clones recovered the phenotype. The gMFI representative histograms are shown in Figs 2B and S2B. Representative images of LDs obtained using confocal microscopy are shown in Figs 2C and S2C.

LDK deletion affected biogenesis and maintenance of LDs in L. infantum

Because our results indicated that LDK deletion leads to a reduction in LD production during the stationary growth phase of L. infantum, we aimed to investigate how LDK-knockout parasites respond in the presence of a substance that stimulates LD production. To achieve this, we incubated both the Cas9-expressing control and mutant parasites with 1.5 µM myriocin, a specific inhibitor of the serine palmitoyltransferase enzyme and an inducer of LD production [18], and then assessed LD production. This concentration has previously been utilized in assays involving T. brucei [6,18]. Notably, the 1.5 μ M myriocin treatment did not result in a reduced growth rate or cell viability for any of the parasites during the evaluated period. Furthermore, myriocin treatment increased LD production in both the Cas9-expressing controls and LDK-knockout clones during both growth phases, compared to the untreated conditions (Fig 3). In the logarithmic growth phase, the geometric mean fluorescence intensity (gMFI) of Nile red-stained LDs was 57.2 and 318.1 for Cas9-expressing controls, and 54.6 and 252.6 for Δ LDK cl1/cl2 parasites, in the presence and absence of myriocin, respectively. In the stationary phase, the gMFI of Nile red-stained LDs was 63.0 and 128.8 for Cas9-expressing controls, and 28.9 and 92.3 for Δ LDK cl1/cl2 parasites, in the presence and absence of myriocin, respectively. A significant reduction in the amount of LDs was observed in *LDK*-knockout clones during both growth phases when compared to the Cas9-expressing controls in the presence of myriocin (Fig 3). This reduction in LDs in the stationary growth phase of LDK-knockout clones was consistent whether myriocin was present or absent (Fig 3). This observation is noteworthy, as without myriocin, LDK-knockout clones exhibited a reduction in LD abundance only during the stationary phase (Figs 2A and 3 and S2A).

LDK deletion altered the intracellular proliferation profile

We investigated the effect of *LDK* deletion in *L. infantum* on the infection profile and intracellular parasite proliferation in THP-1 macrophages. After 3 h of infection, there was no difference in the percentage of infected macrophages nor in the number of intracellular amastigotes. However, after 72 h of infection, we verified that *LDK*-knockout parasites had a reduced capacity to infect macrophages and a reduced number of intracellular amastigotes compared to the Cas9-expressing controls (Fig 4A and 4B, respectively). Reintroduction of the *LDK* gene into the *LDK*-knockout clones recovered the phenotype (Fig 4A and 4B).







Lipid droplet intensity (gMFI Nile red)



Fig 2. Quantification of lipid droplets of Cas9, *LDK* knockouts, and add-back parasites. (A) The graphs represent the mean geometric fluorescence intensity (gMFI) of Nile red. (B) Representative histograms of gMFI of the Nile red. (C) Representative images of lipid droplets obtained using confocal microscopy. Two-way ANOVA with Dunnett's post hoc test was used to compare Cas9-expressing controls and *LDK*-knockouts for each growth phase. *represents significant differences between Cas9 and Δ LDK cl1 and Δ LDK cl2 knockouts (**** p < 0.0001). Bars: 10 µm.

https://doi.org/10.1371/journal.pntd.0011880.g002





https://doi.org/10.1371/journal.pntd.0011880.g003

LDK deletion increased resistance to Sb^{III}, but did not change the susceptibility to amphotericin B, miltefosine, and menadione

We evaluated the susceptibility of both Cas9 and *LDK*-knockout parasites to drugs currently used for leishmaniasis treatment, including antimony, amphotericin B, and miltefosine (Fig



Fig 4. Analysis of the infectivity of Cas9-expressing, Δ LDK null mutants, and add-back parasites in THP-1 macrophages. (A) Percentage of infected macrophages after 3 and 72 h of infection. (B) Number of intracellular amastigotes per 100 macrophages after 3 and 72 h of infection. Ordinary two-way ANOVA test with Dunnett's post hoc test was used to compare Cas9-expressing, Δ LDK null mutants, and add-back parasites at each time point. *represents significant differences in relation to the Cas9-expressing control (** p < 0.01; *** p < 0.001).

https://doi.org/10.1371/journal.pntd.0011880.g004





D



Fig 5. *In vitro* susceptibility assessment of Cas9-expressing, *LDK* knockout, and add-back parasites. Parasites were cultured in the presence of different concentrations of (A) potassium antimony tartrate (Sb^{III}) (50 to 250 μ M), (B) amphotericin B (0.05 to 0.6 μ M), (C) miltefosine (5 to 40 μ M), and (D) menadione (1.0 to 8 μ M). Parasite growth was determined after 48 h of incubation in the presence or absence of drugs. Data were plotted on dose-response curves in which the mean with standard deviations (SD) of three independent experiments performed in triplicate are represented. The EC₅₀ was determined using the linear interpolation method. Two-way ANOVA with post hoc Dunnett's test was applied to compare Cas9-expressing, *LDK*-knockouts, and add-back parasites for each drug concentration. Differences between Cas9-expressing controls and Δ LDK cl2 (* p < 0.05, ** p < 0.01; **** p < 0.001; **** p < 0.001; **** p < 0.001; ****

https://doi.org/10.1371/journal.pntd.0011880.g005

5A-5C). Additionally, we investigated the impact of menadione [21], a compound that induces oxidative stress by increasing peroxide and superoxide radical levels (Fig 5D). Deletion of LDK resulted in increased resistance of L. infantum to Sb^{III} (EC₅₀: ΔLDK cl1: 144.50 μM and Δ LDK cl2: 161.50 μ M), representing a 1.7 and 1.8 -fold increase in resistance for Δ LDK cl1 and Δ LDK cl2, respectively, compared to the Cas9-expressing control (EC₅₀: 87.46 μ M). Notably, the reintroduction of the LDK gene in the LDK-knockout clones restored susceptibility to Sb^{III}, as the EC₅₀ for add-back parasites (cl1 + add-back: 95.24 μ M and cl2 + add-back: 94.14 μ M) was similar to that of the Cas9-expressing control (Fig 5A). However, the deletion of LDK did not affect the susceptibility of the parasites to amphoteric n B, with EC_{50} values ranging from 0.10 to 0.14 μ M for both the Cas9-expressing control and Δ LDK mutant clones (Fig 5B). Similarly, miltefosine susceptibility remained unaltered by LDK deletion, as EC_{50} values for both the Cas9-expressing controls and Δ LDK mutant clones ranged from 10.29 to $13.15 \,\mu$ M (Fig 5C). Additionally, *LDK*-knockout parasites exhibited similar tolerance to oxidative stress induced by menadione (EC_{50}: ΔLDK cl1: 2.60 μM and ΔLDK cl2: 2.99 $\mu M)$ compared to the Cas9-expressing control (EC₅₀: Cas9: 3.27 µM) (Fig 5D). No statistically significant differences were observed between the Cas9-expressing control and LDK-knockout L. infantum at any tested concentrations of amphotericin B, miltefosine, and menadione.

Discussion

Lipid droplets (LDs) are organelles with critical roles in various cellular processes, including lipid metabolism, energy homeostasis, cell signaling, membrane trafficking, inflammation, and innate immunity [10,11,22]. The formation, composition, and functions of LDs are tightly regulated and can vary depending on stimuli, cell types, and the inflammatory environment [11]. LDs contain numerous associated proteins [7,11,22], and in trypanosomatids, two enzymes, TbLpn [12] and LDK [6], have been identified as involved in LD production. In this study, we provide the first evidence of the impact of *LDK* gene deletion using CRISPR/Cas9 on the phenotype of *L. infantum*.

We successfully generated *LDK*-null mutants through CRISPR/Cas9 technology and reintroduced the *LDK* gene into add-back parasites, resulting in high levels of *LDK* transcripts in these parasites. Trypanosomatids have unique molecular characteristics, such as polycistronic transcription, mRNA maturation via trans-splicing, and post-transcriptional regulation of gene expression [23]. The untranslated regions (UTRs) play a crucial role in gene expression by influencing mRNA processing, translation, and degradation [24,25]. Our study employed exogenous UTRs flanking the *LDK* gene in add-back parasites. However, it remains uncertain whether the increased mRNA transcript levels lead to elevated protein levels due to post-transcriptional regulation. Our phenotypic assessment indicated strikingly similar results between add-back and Cas9 parasites, suggesting no substantial difference in LDK protein levels between these groups.

Our data indicate that LDK is not essential for *L. infantum*, as *LDK*-knockout parasites exhibited no alterations in growth patterns, mirroring the behavior of Cas9-expressing controls. These results align with a previous study in which a reduction in *LDK* expression via RNA interference in *T. brucei* did not impact the growth rate [6].

We employed flow cytometry, using both Nile red and Bodipy 493/503, to quantify LDs and lipid levels, as this technique has been utilized in various cell types [26-30]. In trypanosomatids, it has also been employed to quantify LDs [31,32]. Our study successfully employed these methodologies to quantify LDs.

LDs are associated with virulence in several infectious diseases, including *Toxoplasma gondii* and *Plasmodium falciparum* [33]. Similarly, LDs have been linked to virulence in *L*. *infantum*, with their numbers increasing during metacyclogenesis and in intracellular amastigotes [34]. In our study, *LDK* deletion in *L. infantum* reduced the amount of LDs during the stationary phase, although complete depletion did not occur. This finding is consistent with previous research that reported reduced LD abundance when *LDK* expression was diminished in *T. brucei* [6], suggesting that LDK is essential for maintaining normal LD levels in *L. infantum*.

sCells tend to synthesize more LDs in response to increased fatty acids in the medium or lipid homeostasis imbalances [6]. Therefore, we investigated whether *LDK*-knockout parasites would still respond to stimuli that typically induce LD biogenesis. We treated parasites with myriocin, a specific inhibitor of the serine palmitoyltransferase enzyme, a key player in sphingo-lipid metabolism [18,35], known to induce LD production [18]. Our results demonstrated that treatment with 1.5 μ M myriocin for 24 h increased LD production in both Cas9-expressing control and *LDK*-knockout clones, compared to untreated parasites. Importantly, *LDK*-knockout clones exhibited a considerable reduction in LD amount in both growth phases compared to the Cas9-expressing controls in the presence of myriocin. This suggests that LDK deletion may cause changes in other enzymes or pathways indirectly involved in LD biogenesis.

Interestingly, the quantity of LDs in protozoan parasites has been linked to their virulence. In *L. infantum*, LDs play a significant role in infection, as they are responsible for producing prostaglandins like prostaglandin F2 α , which is associated with infection and the enhanced survival of parasites within macrophages [34]. Our study indicated that LDK deletion in *L. infantum* resulted in a reduced number of LDs during the stationary phase. Consequently, we examined its impact on THP-1 macrophage infection and observed that *LDK*-knockout parasites exhibited a decreased ability to maintain infection in macrophages, with a lower intracellular multiplication rate compared to control parasites 72 h post-infection.

Furthermore, we found that *LDK*-knockout increased parasite resistance to Sb^{III}. Lipidomic studies have demonstrated that resistance to antimonials in *Leishmania* is linked to altered lipid composition, including increased levels of very-long-chain fatty acids and ergosterol, suggesting modifications in membrane composition in resistant parasites [36,37]. Our results suggest that LDK deletion may alter the metabolism of other lipids, contributing to the observed resistant phenotype. No significant changes were observed in the susceptibility of *LDK*-knockout parasites to amphotericin B and miltefosine. Resistance of *Leishmania* promastigotes to amphotericin B is characterized by alterations in membrane composition and fluidity, along with changes in the machinery for scavenging reactive oxygen species [38,39]. Although miltefosine is believed to affect lipid metabolism, miltefosine-resistant *Leishmania* parasites did not exhibit significant changes in lipid profiles, suggesting that miltefosine resistance could be attributed to distinct biological processes unrelated to lipid biosynthesis [36]. Additionally, LDK deletion did not alter the mechanisms involved in oxidative stress, as there was no difference in parasite viability after treatment with menadione, a compound that generates oxidative stress by increasing peroxide and superoxide radical levels [21].

In conclusion, our study demonstrated that LDK deletion reduces LD production in *L*. *infantum* during the stationary phase. Although treatment with myriocin induced LD production, *LDK*-knockout parasites exhibited reduced LD amount in both the logarithmic and stationary growth phases compared to control parasites. *LDK*-knockout mutants showed a decreased ability to maintain infection in macrophages and were more resistant to Sb^{III}, suggesting that this deletion may promote changes in pathways related to Sb^{III} metabolism. The biogenesis of lipid droplets in trypanosomatids, notably in *Leishmania*, is not fully characterized, and is a process that suggests the involvement of several enzymes, related to different pathways. These findings suggest that LDK is crucial for maintaining normal LD levels in *L*. *infantum*.

Supporting information

S1 Table. List of primers used in this study. (DOCX)

S1 Fig. Growth curve of promastigote forms of Δ LDK mutant clones. An initial inoculum of 1 x 10⁶ parasites per mL was prepared for the Cas9 parasites, Δ LDK mutant clones cl1 and cl2 and add-back parasites, which were counted every 24 h using the Z1 Coulter Counter. Data represent the mean of three independent experiments performed in triplicates. (TIF)

S2 Fig. Quantification of lipid droplets of Cas9, *LDK*-knockouts, and add-back parasites. (A) The graphs represent the mean geometric fluorescence intensity (gMFI) of Bodipy 493/ 503. (B) Representative histograms of gMFI of the Bodipy 493/503. (C) Representative images of lipid droplets obtained by confocal microscopy. Two-way ANOVA with Dunnett's post hoc test was applied to compare Cas9-expressing controls and *LDK*-knockouts for each growth phase. *represents significant differences between Cas9 and Δ LDK cl1 and Δ LDK cl2 knockouts (** p < 0.01). Bars: 10 µm. (TIF)

Acknowledgments

We thank the Graduate Program in Health Science (Institute René Rachou - IRR/FIOCRUZ) for its support and the Program for Technological Development in Tools for Health- PDTIS-Fiocruz for the use of its facilities at the IRR/FIOCRUZ. The authors also thank Dr. Eva Gluenz for kindly providing plasmids for CRISPR Cas9 genome editing in Leishmania, and Dr. Stephen Beverley for kindly providing the pIR1-SAT expression vector. We would like to thank the Centro de Aquisição e Processamento de Imagens (CAPI -ICB/UMFG) for using the microscope and the Editage (www.editage.com) for English language editing.

Author Contributions

- **Conceptualization:** Juliana Martins Ribeiro, Paula Alves Silva, Ana Maria Murta Santi, Silvane Maria Fonseca Murta.
- **Data curation:** Juliana Martins Ribeiro, Paula Alves Silva, Héllida Marina Costa-Silva, Ana Maria Murta Santi, Silvane Maria Fonseca Murta.
- **Formal analysis:** Juliana Martins Ribeiro, Paula Alves Silva, Héllida Marina Costa-Silva, Ana Maria Murta Santi, Silvane Maria Fonseca Murta.
- Funding acquisition: Silvane Maria Fonseca Murta.
- **Investigation:** Juliana Martins Ribeiro, Paula Alves Silva, Héllida Marina Costa-Silva, Ana Maria Murta Santi, Silvane Maria Fonseca Murta.
- Methodology: Juliana Martins Ribeiro, Paula Alves Silva, Héllida Marina Costa-Silva, Ana Maria Murta Santi, Silvane Maria Fonseca Murta.

Project administration: Silvane Maria Fonseca Murta.

Supervision: Silvane Maria Fonseca Murta.

Validation: Juliana Martins Ribeiro, Paula Alves Silva, Héllida Marina Costa-Silva, Ana Maria Murta Santi, Silvane Maria Fonseca Murta.

Visualization: Silvane Maria Fonseca Murta.

Writing – original draft: Juliana Martins Ribeiro, Paula Alves Silva, Silvane Maria Fonseca Murta.

Writing - review & editing: Juliana Martins Ribeiro, Paula Alves Silva,

Héllida Marina Costa-Silva, Ana Maria Murta Santi, Silvane Maria Fonseca Murta.

References

- 1. WHO. Leishmaniasis. https://www.who.int/health-topics/leishmaniasis#tab=tab_1. Accessed 08 September 2023.
- Alvar J, Vélez ID, Bern C, Herrero M, Desjeux P, Cano J, et al. Leishmaniasis Worldwide and Global Estimates Its Incidence. Plos One. 2012; 7(5):e35671. https://doi.org/10.1371/journal.pone.0035671 PMID: 22693548
- Burza S, Croft SL, Boelaert M. Leishmaniasis. Lancet. 2018; 392(10151):951–970. https://doi.org/10. 1016/S0140-6736(18)31204-2 PMID: 30126638
- Guerin PJ, Olliaro P, Sundar S, Boelaert M, Croft SL, Desjeux P, et al. Visceral leishmaniasis: current status of control, diagnosis, and treatment, and a proposed research and development agenda. Lancet Infect Dis. 2002; 2:494–501. https://doi.org/10.1016/s1473-3099(02)00347-x PMID: 12150849
- Ponte-Sucre A, Gamarro F, Dujardin JC, Barrett MP, López-Vélez R, García-Hernández R, et al. Drug resistance and treatment failure in leishmaniasis: A 21st century challenge. PLoS Negl Trop Dis. 2017; 14;11(12):e0006052. https://doi.org/10.1371/journal.pntd.0006052 PMID: 29240765
- Flaspohler JA, Jensen BC, Saveria T, Kifer CT. Parsons M. A novel protein kinase localized to lipid droplets is required for droplet biogenesis in trypanosomes. Eukaryot Cell. 2010; 9(11):1702–10. https://doi.org/10.1128/EC.00106-10 PMID: 20833891
- Olzmann JA, Carvalho P. Dynamics and functions of lipid droplets. Nat Rev Mol Cell Biol. 2019; 20 (3):137–155. https://doi.org/10.1038/s41580-018-0085-z PMID: 30523332
- Rabhi S, Rabhi I, Trentin B, Piquemal D, Regnault B, Goyard S, et al. Lipid droplet formation, their localization and dynamics during *Leishmania major* macrophage infection. PLoS One. 2016; 12; 11(2): e0148640. https://doi.org/10.1371/journal.pone.0148640 PMID: 26871576
- Bozza P, D'Avila H, Almeida P, Magalhães K, Almeida C, Maya-Monteiro C. Lipid droplets in host–pathogen interactions. Clin Lipidol. 2009; 4(6): 791–807. https://doi.org/10.2217/clp.09.63
- Vallochi AL, Teixeira L, Oliveira KDS, Maya-Monteiro CM, Bozza PT. Lipid droplet, a key player in hostparasite interactions. Front Immunol. 2018; 23;9:1022. <u>https://doi.org/10.3389/fimmu.2018.01022</u> PMID: 29875768
- Tavares VS, de Castro MV, Souza RDSO, Gonçalves IKA, Lima JB, Borges VM, Araújo-Santos T. Lipid droplets of protozoan parasites: survival and pathogenicity. Mem Inst Oswaldo Cruz. 2022; 16;116: e210270. https://doi.org/10.1590/0074-02760210270 PMID: 35195194
- 12. Nejad LD, Serricchio M, Jelk J, Hemphill A, Bütikofer P. TbLpn, a key enzyme in lipid droplet formation and phospholipid metabolism, is essential for mitochondrial integrity and growth of *Trypanosoma brucei*. Mol Microbiol. 2018. https://doi.org/10.1111/mmi.13976 PMID: 29679486
- Aquino GP, Gomes MAM, Salinas RK, Laranjeira-Silva MF. Lipid and fatty acid metabolism in trypanosomatids. Microb Cell. 2021; 8(11):262–275. <u>https://doi.org/10.15698/mic2021.11.764</u> PMID: 34782859
- Halliday C, Billinton K, Wang Z, Madden R, Dean S, Sunter JD, Wheeler RJ. Cellular landmarks of *Try-panosoma brucei* and *Leishmania Mexicana*. Mol Biochem Parasitol. 2019; 230:24–36. https://doi.org/ 10.1016/j.molbiopara.2018.12.003 PMID: 30550896
- Beneke T, Madden R, Makin L, Valli J, Sunter J, Gluenz E. A CRISPR Cas9 high-throughput genome editing toolkit kinetoplastids. R Soc Open Sci. 2017; 4(5): e170095. <u>https://doi.org/10.1098/rsos.</u> 170095 PMID: 28573017
- Robinson KA, Beverley SM. Improvements in transfection efficiency and tests of RNA interference (RNAi) approaches in the protozoan parasite *Leishmania*. Mol Biochem Parasitol. 2003; 128(2):217– 228. https://doi.org/10.1016/s0166-6851(03)00079-3 PMID: 12742588
- Livak KJ, Schmittgen TD. Analysis of relative gene expression data using real-time quantitative PCR and the 2(-Delta Delta C(T)) method. Methods. 2001; 25(4):402–408. <u>https://doi.org/10.1006/meth.</u> 2001.1262 PMID: 11846609
- Fridberg A, Olson CL, Nakayasu ES, Tyler KM, Almeida IC, Engman DM. Sphingolipid synthesis is necessary for kinetoplast segregation and cytokinesis in *Trypanosoma brucei*. J Cell Sci. 2008; 15; 121(Pt 4):522–35. https://doi.org/10.1242/jcs.016741 PMID: 18230649

- Melo RCN, D'Avila H, Wan H-C, Bozza PT, Dvorak AM, Weller PF. Lipid bodies in inflammatory cells: structure, function, and current imaging techniques. J Histochem Cytochem Off J Histochem Soc. 2011; 59(5):540–56. https://doi.org/10.1369/0022155411404073 PMID: 21430261
- Huber W, Koella JC. A comparison of three methods of estimating EC50 in studies of drug resistance of malaria parasites. Acta Trop. 1993; 55:257–61. https://doi.org/10.1016/0001-706x(93)90083-n PMID: 8147282
- Sakagami H, Satoh K, Hakeda YKM. Apoptosis-inducing activity of vitamin C and vitamin K. Cell Mol Biol. 2000; 46:129–43. PMID: 10726979
- Onal G, Kutlu O, Gozuacik D, Emre SD. Lipid droplets in health and disease. Lipids Health Dis. 2017; 16 (128). https://doi.org/10.1186/s12944-017-0521-7 PMID: 28662670
- Maslov DA, Opperdoes FR, Kostygov AY, Hashimi H, Lukeš J, Yurchenko V. Recent advances in trypanosomatid research: genome organization, expression, metabolism, taxonomy and evolution. Parasitol. 2019; 146(1):1–27. https://doi.org/10.1017/S0031182018000951 PMID: 29898792
- Clayton C. Regulation of gene expression in trypanosomatids: living with polycistronic transcription. Open Biol. 2019; 9(6):190072. https://doi.org/10.1098/rsob.190072 PMID: 31164043
- Clayton CE. Gene expression in Kinetoplastids. Curr Opin Microbiol. 2016; 32:46–51. https://doi.org/10. 1016/j.mib.2016.04.018 PMID: 27177350
- Greenspan P, Mayer EP, Fowler SD. Nile red: a selective fluorescent stain for intracellular lipid droplets. J Cell Biol. 1985; 100(3):965–73. https://doi.org/10.1083/jcb.100.3.965 PMID: 3972906.
- Boumelhem BB, Pilgrim C, Zwicker VE, Kolanowski JL, Yeo JH, Jolliffe KA, New EJ, Day ML, Assinder SJ, Fraser ST. Intracellular flow cytometric lipid analysis - a multiparametric system to assess distinct lipid classes in live cells. J Cell Sci. 2022; 135(5):jcs258322. https://doi.org/10.1242/jcs.258322 PMID: 34114626
- Rumin J, Bonnefond H, Saint-Jean B, Rouxel C, Sciandra A, Bernard O, Cadoret JP, Bougaran G. The use of fluorescent Nile red and BODIPY for lipid measurement in microalgae. Biotechnol Biofuels. 2015; 8:42. https://doi.org/10.1186/s13068-015-0220-4 PMID: 25788982
- Wolins NE, DeHaan KN, Cifarelli V, Stoeckman AK. Normalized neutral lipid quantitation by flow cytometry. J Lipid Res. 2018; 59(7):1294–1300. https://doi.org/10.1194/jlr.D084871 PMID: 29764924
- Hagberg CE, Li Q, Kutschke M, Bhowmick D, Kiss E, Shabalina IG, Harms MJ, Shilkova O, Kozina V, Nedergaard J, Boucher J, Thorell A, Spalding KL. Flow Cytometry of Mouse and Human Adipocytes for the Analysis of Browning and Cellular Heterogeneity. Cell Rep. 2018; 24(10):2746–2756.e5. <u>https://doi.org/10.1016/j.celrep.2018.08.006</u> PMID: 30184507
- Santi AMM, Ribeiro JM, Reis-Cunha JL, Burle-Caldas GA, Santos IFM, Silva PA, Resende DM, Bartholomeu DC, Teixeira SMR, Murta SMF. Disruption of multiple copies of the Prostaglandin F2alpha synthase gene affects oxidative stress response and infectivity in *Trypanosoma cruzi*. PLoS Negl Trop Dis. 2022; 16(10):e0010845. https://doi.org/10.1371/journal.pntd.0010845 PMID: 36260546
- Basselin M, Robert-Gero M. Alterations in membrane fluidity, lipid metabolism, mitochondrial activity, and lipophosphoglycan expression in pentamidine-resistant *Leishmania*. Parasitol Res. 1998; 84 (1):78–83. https://doi.org/10.1007/s004360050361 PMID: 9491432
- Saka HA, Valdivia R. Emerging roles for lipid droplets in immunity and host-pathogen interactions. Annu Rev Cell Dev Biol. 2012; 28:411–37. https://doi.org/10.1146/annurev-cellbio-092910-153958 PMID: 22578141
- 34. Araújo-Santos T, Rodríguez NE, Moura-Pontes S, Dixt UG, Abánades DR, Bozza PT, et al. Role of prostaglandin F2α production in lipid bodies from *Leishmania infantum chagasi*: insights on virulence. J Infect Dis. 2014; 210(12):1951–61. https://doi.org/10.1093/infdis/jiu299 PMID: 24850789
- Zhang K, Showalter M, Revollo J, Hsu FF, Turk J, Beverley SM. Sphingolipids are essential for differentiation but not growth in *Leishmania*. EMBO J. 2003; 22(22):6016–6026. https://doi.org/10.1093/emboj/ cdg584 PMID: 14609948
- **36.** Biagiotti M, Dominguez S, Yamout N, Zufferey R. Lipidomics and anti-trypanosomatid chemotherapy. Clin Transl Med. 2017; 6(1):27. https://doi.org/10.1186/s40169-017-0160-7 PMID: 28766182
- Mathur R, Das R, Ranjan A, Shah C. Elevated ergosterol protects *Leishmania* parasites against antimony-generated stress. FASEB J. 2015; 29(10):4201–13. <u>https://doi.org/10.1096/fj.15-272757</u> PMID: 26116701
- Purkait B, Kumar A, Nandi N, Sardar AH, Das S, Kumar S, et al. Mechanism of amphotericin B resistance in clinical isolates of *Leishmania donovani*. Antimicrob Agents Chemother. 2012; 56(2):1031–41. https://doi.org/10.1128/AAC.00030-11 PMID: 22123699
- Mbongo N, Loiseau PM, Billion MA, Robert-Gero M. Mechanism of amphotericin B resistance in *Leishmania donovani* promastigotes. Antimicrob Agents Chemother. 1998; 42(2):352–7. https://doi.org/10. 1128/AAC.42.2.352 PMID: 9527785